Author’s response to reviews

**Title:** Diagnostic accuracy of point-of-care testing for acute coronary syndromes, heart failure and thromboembolic events in primary care: a cluster-randomised controlled trial

**Authors:**
- Yuki Tomonaga (yukit@access.uzh.ch)
- Felix Gutzwiller (gutzwil@ifspm.uzh.ch)
- Thomas F Lüscher (karlue@usz.uzh.ch)
- Walter F Riesen (Walter.Riesen@ikch.ch)
- Markus Hug (markus.hug@hin.ch)
- Albert Diemand (a-l.diemand@bluewin.ch)
- Matthias Schwenkglenks (matthias.schwenkglenks@uzh.ch)
- Thomas D Szucs (thomas.szucs@unibas.ch)

**Version:** 4 **Date:** 9 March 2011

**Author’s response to reviews:** see over
Zurich, 09 March 2011

MS: 6806930873847505 - Clinical benefit of point-of-care testing for acute coronary syndromes, heart failure and thromboembolic events in primary care: a cluster-randomised controlled trial.

Dear Editors,

Thank you for accepting our manuscript. As requested, we finally inserted the required trial registration number at the foot of the abstract.

Sincerely

University of Zurich
Institute of Social and Preventive Medicine

Yuki Tomonaga
Research associate

Thomas D. Szucs
Prof. Dr. med.